Author's response to reviews

Title: Patterns of delays amongst pulmonary tuberculosis patients in Lagos, Nigeria.

Authors:

Dr Olumuyiwa O Odusanya (olumuyiwa88@hotmail.com)
Joseph O Babafemi (oluuyiwa88@hotmail.com)

Version: 4 Date: 20 May 2004

18th May, 2004
The Editor,
BMC Public Health

Dear Sirs,

Re: Patterns of delays amongst pulmonary tuberculosis patients in Lagos, Nigeria

We the undersigned authors wish to submit the final copy of the above mentioned manuscript. The only change made to the present text is in the discussion on page 8, where it was stated that these results confirm the findings of Enwuru et al. We have formatted the text in keeping with the Journal style. We hope the manuscript is now publishable.

Yours sincerely

1. Dr. Odusanya O.O.
Department of Community Health & Primary Care
Lagos State University College of Medicine
P.M.B. 21266,

2. Dr. Babafemi J.O.
Mainland Hospital
Yaba, Lagos, Nigeria